Author’s response to reviews

Title: How can educators improve the perception of happiness for pre-clinical medical students?

Authors:

Sami Ahmad Ghani (sami_ghani@hotmail.com)
Muhammed Aizaz us Salam (aizazsalam@yahoo.com)
George Chukwuemeka Oyekwe (goyekwe@gmail.com)
Sharfraz Riaz Choudhury (sharfrazc@outlook.com)

Version: 2 Date: 26 Aug 2020

Author’s response to reviews:

Several minor changes have been made to the sentence structure and wording throughout the manuscript. Grammatical errors have been corrected. This revised manuscript has been proof read by all authors. As requested, a single, clean version has been uploaded with no highlighted changes.